Author’s response to reviews

Title: Formulation of Evidence-Based Messages to Promote the Use of Physical Activity to Prevent and Manage Alzheimer's Disease

Authors:

Kathleen Martin Ginis (kathleen_martin.ginis@ubc.ca)
Jennifer Heisz (heiszjj@mcmaster.ca)
John Spence (jc.spence@ualberta.ca)
Ilana Clark (clarkib@mcmaster.ca)
Jordan Antflick (jantflick@braininstitute.ca)
Chris Arden (carden@yorku.ca)
Christa Costas-Bradstreet (ccbradstreet@participaction.com)
Mary Duggan (mduggan@csep.ca)
Audrey Hicks (hicksal@mcmaster.ca)
Amy Latimer-Cheung (amy.latimer@queensu.ca)
Laura Middleton (lmiddlet@uwaterloo.ca)
Kirk Nylen (knynlen@braininstitute.ca)
Donald Paterson (dpaterso@uwo.ca)
Michael Rotondi (mrotondi@yorku.ca)

Version: 2 Date: 25 Jan 2017

Author’s response to reviews:

Thank you for your favourable response to our revised manuscript. Below, please find my responses to the two outstanding items you noted in your recent email:

1. Consent to Publish

Please provide a Consent to publish statement for Katherine Rankin, who is explicitly named as part of the Expert panel in Table 1. but is not listed as an author of the manuscript.
I have obtained Consents to publish from both Ms. Rankin and from Hans Messersmith who was also part of the Expert panel, but not part of the authorship on this paper. These forms are now uploaded and this consent process is now mentioned in the first part of the Declarations section on page 30 (under 'Consent to Publish') as follows:

All authors have consented to publish this manuscript. Two expert panel members (HM, KR) did not make authorship contributions to the manuscript. They have provided written, signed consent for their names to be published in Table 1.

2. Authorship

Please ensure that the individual contributions of each of the authors is explicitly mentioned in this section.

In the 'Authors' contributions' section of the Declarations page on page 30, it is now explicitly mentioned that: "KMG, JH, JCS, IBC, CIA, CC-B, MD, ALH, AEL-C, LM, KN, DHP, CP, and MR participated in the expert panel meeting, provided feedback on manuscript drafts, and read and approved the final version of the manuscript."